Author’s response to reviews

Title: The occurrence of comorbidities to affective and anxiety disorders among older people with intellectual disability in comparison with the general population: A register study

Authors:

Nadia El Mrayyan (nadia.el_mrayyan@med.lu.se)
Jonas Eberhard (jonas.eberhard@med.lu.se)
Gerd Ahlström (Gerd.Ahlstrom@med.lu.se)

Version: 1 Date: 16 Feb 2019

Author’s response to reviews:

Dear Editor,

Thank you very much for your review. We appreciate the comments. Kindly find attached a new version of the revised manuscript according with your two corrections. We changed the “mental retardation” term to “Intellectual disability” in the keywords. However, it in the methods section we kept “mental retardation” because it’s the diagnosis term used in the ICD-10 classification for Intellectual disability. In addition, the other two terms are used in published articles therefore it’s not possible to change them in reference list. Finally, the correction regards the figure is revised as you requested.

Sincerely,

Nadia El Mrayyan on behalf of all authors.